Author’s response to reviews

Title: Modeling community integration in workers with delayed recovery from mild traumatic brain injury

Authors:

Tatyana Mollayeva (tatyana.mollayeva@utoronto.ca)
Colin M Shapiro (colinshapiro@rogers.com)
Shirin Mollayeva (shirin.mollayeva@utoronto.ca)
David Cassidy J (dcassidy@health.sdu.dk)
Angela Colantonio (angela.colantonio@utoronto.ca)

Version: 6 Date: 28 August 2015

Author’s response to reviews: see over
August 27, 2015

Dear Professor Liming Li and the editorial team,

My co-authors and I are grateful that our work is now considered for publication in the *BMC Neurology*. The updated version of the manuscript follows the journal’s guidelines.

Sincerely,

Tatyana

Tatyana Mollayeva, MD, PhD
Toronto Rehabilitation Institute-University Health Network
550 University Avenue, Rm 11207
Toronto, Ontario M5G 2A2
Email: tatyana.mollayeva@utoronto.ca
Tel: 416-597-3422 ext 7848

- Please provide in the title page the email addresses of all authors
  
  **Response:** this has been done

- Please detail funding in the Acknowledgements and remove the section Funding from the manuscript
  
  **Response:** this has been done

- Please clarify whether informed consent obtained from participants was verbal or written
  
  **Response:** we obtained written consent from each participant. We have added this information to the manuscript.

- Please rename section Interpretation as Discussion
  
  **Response:** this has been done

- Please format the Authors Contributions according to our Instructions for Authors
  
  **Response:** this has been done

- Please place figure titles/legends at the end of the main manuscript and not within the figure files themselves
  
  **Response:** this has been done